Author's response to reviews

**Title:** The potential impact of plain packaging of cigarette products among Brazilian young women: an experimental study

**Authors:**

Christine M White (c5white@uwaterloo.ca)
David Hammond (dhammond@uwaterloo.ca)
James F Thrasher (thrasher@mailbox.sc.edu)
Geoffrey T Fong (gfong@uwaterloo.ca)

**Version:** 2 **Date:** 20 December 2011

**Author's response to reviews:**

To Whom it May Concern,

Thank you for your initial review feedback. As requested, the following changes have been made to our manuscript:

1. Added clarification that the study received ethics clearance from the Office of Research Ethics at the University of Waterloo, Ontario, Canada (page 3, paragraph 1).

Thank you for considering our manuscript.

Sincerely,

David Hammond